\documentclass[10pt, onehalfspace,twoside,reqno]{article}
%\documentclass[twoside,reqno,11pt]{fcaa}
%\input fcaa_style
\usepackage{hyperref}
 \setcounter{page}{1}
  \thispagestyle{empty}
 \def\theequation{\arabic{section}.\arabic{equation}}
 %\def\themycopyrightfootnote{\vspace*{3pt}

%\usepackage{latexsym}
\usepackage{graphicx}
\usepackage{epsfig}
%\usepackage{amsthm}
\usepackage{fancyhdr}
\usepackage{latexsym}
\usepackage{amsfonts}
\usepackage{amssymb}




\usepackage[T1]{fontenc}
\usepackage{geometry}
\usepackage[latin1]{inputenc}
\usepackage{amssymb,amsmath,mathrsfs}

\usepackage[frenchb,english]{babel}




\setcounter{MaxMatrixCols}{10}














%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\newcommand{\R}{\mathbb{R}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\M}{\mathbb{M}}
\newcommand{\C}{\ensuremath{\mathscr{C}}}
\newcommand{\D}{\ensuremath{\mathscr{D}}}
\newcommand{\U}{\ensuremath{\mathscr{U}}}
\newcommand{\V}{\ensuremath{\mathscr{V}}}
\newcommand{\Ww}{\ensuremath{\mathscr{W}}}
\newcommand{\Aa}{\ensuremath{\mathscr{A}}}
\newcommand{\Bb}{\ensuremath{\mathscr{B}}}
\newcommand{\F}{\ensuremath{\mathscr{F}}}


\def\tscale {\rightharpoonup\rightharpoonup}
\def\cf {\rightharpoonup}
\def\var {\varepsilon}
\def \display {\displaystyle}



%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%
%\theoremstyle{defi}
\newtheorem{definition}{Definition}[section]
%\newtheorem{remark}{Remark}[section]
%\newtheorem{example}{Example}[section]


%\theoremstyle{theorem}
\newtheorem{lemma}{Lemma}[section]
\newtheorem{theorem}{Theorem}[section]
%\newtheorem{corollary}{Corollary}[section]
\newtheorem{example}{Example}[section]
%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%




 \def\proofname{\indent\rm P\ r\ o\ o\ f.}
 \def\proofend{\hfill$\Box$}
 % or the same:
 \def\qed{\hfill$\Box$} % instead of \square






\newtheorem{conseq}{\textsc{Consequence}}
%\newtheorem{lem}{\textsc{Lemma}}
%\newtheorem{theo}{\textsc{Theorem}}
\newtheorem{prop}{\textsc{Proposition}}
\newtheorem{rem}{\textsc{Remark}}
\newtheorem{cor}{\textsc{Corollary}}
%\newtheorem{definition}{\textsc{Definition}}
\newtheorem{proof}{\textsc{Proof}}
\newtheorem{prel}{\textsc{Preliminaries}}





 \setcounter{page}{1}
  \thispagestyle{empty}







\pagestyle{myheadings}

\markboth{\hfil Nonlinear fractional differential equation \hfil }
{\hfil B. Tellab \&  K. Haouam \hfil}
\begin{document}
\begin{thebibliography}{99}

\bibitem{b01}A.A. Kilbas, H.M. Srivastava and J.J. Trujillo, Theory and Applications of Fractional Differential Equations, volume 204 of North-Holland Mathematics Studies. Elsevier, Amesterdam, 2006.
\bibitem{b02}A.M. El-Sayed and E.O. Bin-Tahar , "Positive Nondecreasing solutions for a Multi-Term Fractional-Order Functional Differential Equation With Integral Conditions", Electronic Journal of Differential Equations, Vol. 2011, No. 166, pp. 18, (2011).
\bibitem{b03}E. Zeidler, Nonlinear functional analysis and its applications Fixed point theorems, Springer-Verlag, New york Berlin Heiderberg, Tokyo 1985.
\bibitem{b04} I. Podlubny,  Fractional differential equations. Mathematics in science and engineering, vol. 198. New York/London:
Springer;1999.
\bibitem{b05} J. K. Hale and S. Verduyn, Introduction to Functional Differential Equations, Applied Mathematical Sciences, 99, Springer-Verlag, New York, 1993.
\bibitem{b06} K. Deng, HA. Levine,  The role of critical exponents in Blow-up
theorems: the sequel, J Math Anal Appl 243(2000) 85-126.
\bibitem{b07}  K. Saoudi, K. Haouam,  Critical exponent for nonlinear hyperbolic
system with spatio-temporal fractional derivatives. International Journal
of Applied Mathematics vol 24, $N^{o}$ 6 2011 pp 861-871.
\bibitem{b08} K. Sotiris Ntouyas, Existence Results For First Order Boundary Value Problems For Fractional Differential equations and Inclusions With Fractional Integral Boundary Conditons, Journal of Fractional Calculus and Applications, Vol. 3 July 2012, No. 9, pp 1-14.
\bibitem{b09}L. Gaul, P. Klein and kempfle, Damping description involving fractional operators, Mech. Systems Signal Proceesing 5 (1991), 81-88.
\bibitem{b10}M. Benchohra and F. Ouaar, Existence Results for nonlinear fractional differential equations with integral boundary conditions, Bullten Of Mathematical analysis and Applications, Vol. 2 issue4, pp. 7-15, (2010).
\bibitem{b11} M. Kirane, N-e. Tatar,  Nonexistence of solutions to a hyperbolic equation with a time fractional damping.
 Z Anal Anwend (J Anal Appl) 2006;25:131-42.
\bibitem{b12}R. Gorenflo, Abel integral equations with special emphasis on applications, Lectures in Mathematical Sciences, Vol. 13, University of Tokyo, 1996.
\bibitem{b13}R. Hilfer, Applications of Fractional Calculus in Physics, World Scientific, Singapore, 2000.
\bibitem{b14}RL. Bagley, PJ. Torvik,  A theoretical basis for the application of fractional calculus to viscoelasticity.
J Rheol 1983;27:201-10
\bibitem{b15}R. P. Agarwal, Y. Zhou and Y. He, Existence of fractional neutral functional differential équations, Comput. Math. Appl. 59 (3) (2010), 1095-1100.
\bibitem{b16}R. W. Ibrahim, Existence and uniquness of holomographyic solutions for fractinal Cauchy problem, J. Math. Anal. Appl. 380, pp. 232-240, (2011).
\bibitem{b17}S. B. Hadid, "Local and global existence theorems on differential equations of non-integer order", J. Fractional Calculus, Vol. 7, pp. 101-105, May (1995).
\bibitem{b18} SG. Samko, AA. Kilbas, Marichev OI. Fractional integrals and derivatives: theory and applications. Amesterdam:
 Gordon and Breach; 1993[Engl. Trans. from the Russian eddition 1987].
\bibitem{b19}S. Zhang, Positive solutions for boundary-value problems of nonlinear fractional equations, Electron. J. Differential Equations 2006, No. 36, 12 pp.
\bibitem{b20}W. G. Glokle and T. F. Nonnenmacher, A fractional calculus approach of self-similar protein dynamics, Biophys. J. 68 (1995), 46-53.
\end{thebibliography}

\title{\textbf{{Existence and uniqueness of solutions for  nonlinear hyperbolic fractional differential equation with integral boundary conditions
}}}


\author{\textbf{Brahim \textsc{Tellab*}} ~ and  ~
\textbf{Kamel \textsc{Haouam**}}}

\date{}

\maketitle

\begin{center}
{\scriptsize {* Departement of Mathematics University Mentouri , } \\[0pt]
Constantine 1, Algeria. \\[0pt]
}

{\scriptsize {** Mathematics and Informatics dept, LAMIS Laboratory, Tebessa
University,} \\[0pt]
12000 Tebessa, Algeria. ~ \\[0pt]
Mail : brahimtel@yahoo.fr (B. Tellab)\\[0pt]
haouam@yahoo.fr (K. Haouam) }
\end{center}

\begin{abstract}
%\bigskip

In this paper, we investigate the existence and uniqueness of solution for second order nonlinear fractional differential equation with integral boundary conditions. Our result is an application of the Banach contraction principle and the Krasnoselskii fixed point theorem.
\medskip

\noindent {\bf Keywords:} fractional derivatives, contraction principle, fixed point theorem, integral equation.

\smallskip

\end{abstract}

\maketitle









\section{Introduction}\label{sec:1}

\medskip

Fractional differential equations have been of great interest and attracted many researchers in recent years, this is due to the development of the above cited concept. It have found applications in several different disciplines as physics, engineering, economics, electrochemistry, electromagnetism etc. (See \cite{b03, b04, b09, b13, b20}).\\
Such equations have recently proved to be valuable tools in modelling of many phenomena. ( See papers \cite{b02,b07,b12, b14, b16,   b17} ).\\
\\In \cite{b10}, Benchohra and Ouaar discussed the existence of solutions to the boundary value problem:


\begin{eqnarray}
\label{equ1}
^CD^{\alpha}y(t)=f(t,y(t)),\hspace{0.2cm} t\in J=[0,T],\hspace{0.4cm}\alpha\in(0,1],
\end{eqnarray}
\begin{eqnarray}
\label{equ2}
y(0)+\mu\int_{0}^{T}y(s)ds=y(T),
\end{eqnarray}
 $^CD^{\alpha}$ is the Caputo fractional derivative,  $f:J\times\R\rightarrow\R$ is continuous function and $\mu\in\R^{\ast}.$\\
In \cite{b08}, Sotiris K. Ntouyas investigated the existence and uniqueness of solutions of the following problem:
\begin{eqnarray}
\label{equ3}
^CD^{q}x(t)=f(t,x(t)),\hspace{0.2cm} 0<t<1,\hspace{0.2cm}0<q\leq1,
\end{eqnarray}
\begin{eqnarray}
\label{equ4}
x(0)=\alpha I^{p}x(\eta),\hspace{0.2cm}0<\eta<1,
\end{eqnarray}
$^CD^{q}$ denotes always the Caputo fractional derivative of order $q,$  $f:[0,1]\times\R\rightarrow\R$ is continuous function, $\alpha\in\R$ such that $\alpha\neq\Gamma(p+1)/\eta^{p},$ $\Gamma$ is the Euler funtion and $I^{p},$ 0<p<1 is the  Riemann-Liouville fractional integral of order $p.$\\
In this paper, we consider the following nonlinear fractional differential equation with integral boundary conditions:

\begin{eqnarray}
\label{equ5}
^CD^{\alpha}y(t)=f(t,y(t)),\hspace{0.2cm} t\in J=[0,1],
\end{eqnarray}
\begin{eqnarray}
\label{equ6}
y(0)=\int_{0}^{1}y(s)ds
\end{eqnarray}
\begin{eqnarray}
\label{equ7}
y(1)=\frac{1}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}y(s)ds
\end{eqnarray}
where $^CD^{\alpha}$ is the Caputo fractional derivative of order $\alpha,$ $1<\alpha\leq2,$ $0<\beta\leq1$ and $f:[0,1]\times\R\rightarrow\R$ is continuous function.
\section{preliminaries}\label{sec:2}

Now, we present some basic definitions and lemmas of fractional calculus which will be used in our theorems \cite{b01, b04, b18}.
%%%%  definition 1 %%%%
\begin{definition}\label{Def1}
For a differentiable function $h:[0,+\infty)\rightarrow\R,$ the Caputo derivative of fractional order $\alpha$ is defined by
$$
^CD^{\alpha}h(t)=\frac{1}{\Gamma(n-\alpha)}\int_{0}^{t}(t-s)^{n-\alpha-1}h^{(n)}(s)ds,\hspace{0,3cm}n-1<\alpha<n,n=[\alpha]+1,
$$
where $[\alpha]$ denotes the integer part of $\alpha$ and $\Gamma$ is the gamma function.
\end{definition}
%%%%  definition 2 %%%%
\begin{definition}
  The Riemann-Liouville fractional integral of order $\alpha$ is given by
$$
I^{\alpha}h(t)=\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}h(s)ds,
$$
where $h:\R^{+}\rightarrow\R$ is a Lebesgue measurable function, provided that the integral exists.
\end{definition}
\begin{lemma}\label{lemma1}
\cite{b19} Let $\alpha>0,$ then the differential equation
$$
^CD^{\alpha}h(t)=0
$$
has solutions
$$
h(t)=c_{0}+c_{1}t+c_{2}t^{2}+...+c_{n-1}t^{n-1},
$$
$c_{i}\in\R,$ $i=0,1,2,...,n-1,$ $n=[\alpha]+1.$

\end{lemma}
\begin{lemma}\label{lemma2}
\cite{b19} Let $\alpha>0,$ then
$$
I^{\alpha}{^CD^{\alpha}}h(t)=h(t)+c_{0}+c_{1}t+c_{2}t^{2}+...+c_{n-1}t^{n-1},
$$
where $c_{i}\in\R,$ $i=0,1,2,...,n-1,$ $n=[\alpha]+1.$
\end{lemma}
\begin{lemma}\label{lemma3}
Let $1<\alpha\leq2$ and let $h:J\times\R\rightarrow\R$ be a given continuous function. Then, the boundary-value problem
\begin{eqnarray}
\label{equ8}
^CD^{\alpha}y(t)=h(t),\hspace{0.2cm} t\in J
\end{eqnarray}
\begin{eqnarray}
\label{equ9}
y(0)=\int_{0}^{1}y(s)ds
\end{eqnarray}
\begin{eqnarray}
\label{equ10}
y(1)=\frac{1}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}y(s)ds
\end{eqnarray}
has a unique solution defined by:
\begin{eqnarray}
&&y(t)=\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}h(s)ds+\nonumber\\
&&\int_{0}^{1}\bigg[\frac{1}{\gamma_{1}\Gamma(\alpha)\Gamma(\beta)}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}dr-
\frac{(1-s)^{\alpha-1}}{\gamma_{1}\Gamma(\alpha)}+\bigg(\frac{2\gamma_{2}}{\gamma_{1}\alpha\Gamma(\alpha)}-\frac{2t}
{\alpha\Gamma(\alpha)}
\bigg)(1-s)^{\alpha}\bigg]h(s)ds\nonumber
\end{eqnarray}
where
$$
\gamma_{1}=1-\frac{1}{\Gamma(\beta+1)},\hspace{0.4cm}\gamma_{2}=1-\frac{1}{\Gamma(\beta+2)}.
$$
\end{lemma}
%\begin{proof}
\subsubsection*{ \bf Proof of Lemma~\ref{lemma3}}


Applying lemma \ref{lemma2}, we can reduce the problem (\ref{equ8})-(\ref{equ10})to an equivalent integral equation
\begin{eqnarray}
\label{equ11}
y(t)&=&I_{0}^{\alpha}h(t)+c_{0}+c_{1}t\nonumber\\
&=&\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}h(s)ds+c_{0}+c_{1}t,
\end{eqnarray}
for some constants $c_{0},c_{1}\in\R.$ By integrating and using Fubini's theorem, we obtain:
\begin{eqnarray}
\label{equ12}
\int_{0}^{1}y(s)ds=\int_{0}^{1}\frac{(1-\tau)^{\alpha}}{\alpha\Gamma(\alpha)}h(\tau)d\tau+c_{0}+\frac{c_{1}}{2}.
\end{eqnarray}
Applying (\ref{equ9}), we find:
$$
y(0)=c_{0},
$$
and with (\ref{equ12}), we arrive to
$$
y(0)=\int_{0}^{1}\frac{(1-\tau)^{\alpha}}{\alpha\Gamma(\alpha)}h(\tau)d\tau+c_{0}+\frac{c_{1}}{2},
$$
then,
\begin{eqnarray}
\label{equ13}
c_{1}=-2\int_{0}^{1}\frac{(1-\tau)^{\alpha}}{\alpha\Gamma(\alpha)}h(\tau)d\tau.
\end{eqnarray}
Now, we operate (\ref{equ10}) and (\ref{equ11}), we get
\begin{eqnarray}
\frac{1}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}y(s)ds&=&\frac{1}{\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{0}^{s}
(1-s)^{\beta-1}(s-r)^{\alpha-1}h(r)drds\nonumber\\
&+&\frac{c_{0}}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}ds+\frac{c_{1}}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}sds,\nonumber
\end{eqnarray}
that is to say:
\begin{eqnarray}
\frac{1}{\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}h(s)ds+c_{0}+c_{1}&=&
\frac{1}{\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{0}^{s}
(1-s)^{\beta-1}(s-r)^{\alpha-1}h(r)drds\nonumber\\
&+&\frac{c_{0}}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}ds+\frac{c_{1}}{\Gamma(\beta)}\int_{0}^{1}(1-s)^{\beta-1}sds,\nonumber
\end{eqnarray}
after the simplifications, we obtain:
\begin{eqnarray}
\frac{1}{\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}h(s)ds+c_{0}+c_{1}&=&
\frac{1}{\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{s}^{1}
(1-r)^{\beta-1}(r-s)^{\alpha-1}h(s)drds\nonumber\\
&+&\frac{c_{0}}{\Gamma(\beta+1)}+\frac{c_{1}}{\Gamma(\beta+2)},\nonumber
\end{eqnarray}
which may be written,
\begin{eqnarray}
\label{equ14}
\bigg(1-\frac{1}{\Gamma(\beta+1)}\bigg)c_{0}+\bigg(1-\frac{1}{\Gamma(\beta+2)}\bigg)c_{1}&=&\frac{1}{\Gamma(\alpha)\Gamma(\beta)}
\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}h(s)drds\nonumber\\
&-&\frac{1}{\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}h(s)ds.
\end{eqnarray}
In putting  $\gamma_{1}=1-\frac{1}{\Gamma(\beta+1)},\hspace{0.4cm}\gamma_{2}=1-\frac{1}{\Gamma(\beta+2)},$ (\ref{equ14}) becomes:
\begin{eqnarray}
\gamma_{1}c_{0}+\gamma_{2}c_{1}=\frac{1}{\Gamma(\alpha)\Gamma(\beta)}
\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}h(s)drds
-\frac{1}{\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}h(s)ds.\nonumber
\end{eqnarray}
Using (\ref{equ13}), we obtain:
\begin{eqnarray}
\label{equ15}
c_{0}&=&\frac{1}{\gamma_{1}\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}h(s)drds
-\frac{1}{\gamma_{1}\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}h(s)ds\nonumber\\
&+&\frac{2\gamma_{2}}{\gamma_{1}\alpha\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha}h(s)ds
\end{eqnarray}
a combination of (\ref{equ11}),(\ref{equ13}) and (\ref{equ15}) leads us to:
\begin{eqnarray}
y(t)&=&\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}h(s)ds+\frac{1}{\gamma_{1}\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}
\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}h(s)drds\nonumber\\
&-&\frac{1}{\gamma_{1}\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}h(s)ds+\bigg[\frac{2\gamma_{2}}{\gamma_{1}\alpha\Gamma(\alpha)}
-\frac{2t}{\alpha\Gamma(\alpha)}\bigg]\int_{0}^{1}(1-s)^{\alpha}h(s)ds,\nonumber
\end{eqnarray}
i.e,
\begin{eqnarray}
&&y(t)=\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}h(s)ds+\nonumber\\
&&\int_{0}^{1}\bigg[\frac{1}{\gamma_{1}\Gamma(\alpha)\Gamma(\beta)}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}dr-
\frac{(1-s)^{\alpha-1}}{\gamma_{1}\Gamma(\alpha)}+\bigg(\frac{2\gamma_{2}}{\gamma_{1}\alpha\Gamma(\alpha)}-\frac{2t}
{\alpha\Gamma(\alpha)}
\bigg)(1-s)^{\alpha}\bigg]h(s)ds\nonumber
\end{eqnarray}
{\flushright$\blacksquare$\\}
%\end{proof}
\section{Existence and uniqueness result}\label{sec:3}

\begin{theorem}\label{Th1}{\textbf{( fixed point theorem of Banach)}}\cite{b06}
Let $X$ be a  Banach space and $T:X\longrightarrow X$ a contracting mapping. Then $T$ has a unique fixed point i.e.
$$
\exists! x\in X: \hspace{0.1cm}Tx=x.
$$
\end{theorem}
Our first result of existence is based on theorem of Banach contracting application.
\begin{theorem}\label{Th2}
Suppose that the function $f:[0,1]\times\R\rightarrow\R$ is continuous and there is a constant $L>0$ such that:\\
$(H_{1}): |f(t,x)-f(t,y)|\leq L|x-y|,\hspace{0.2cm}t\in[0,1],\hspace{0.2cm}x,y\in\R.$\\








If $LA<1,$ then the boundary value problem (\ref{equ5})-(\ref{equ7}) has a unique solution, where
\begin{eqnarray}
\label{equ16}
A=\frac{1}{\Gamma(\alpha+1)}+\frac{\mathfrak{B(\beta,\alpha)}}{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+
\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}+\frac{2}{\Gamma(\alpha+2)}.
\end{eqnarray}
\end{theorem}
%\textbf{Proof:}\\
\subsubsection*{ \bf Proof of Theorem~\ref{Th2}}
We define the operator $F,$ by:
\begin{eqnarray}
\label{equ17}
&&(Fy)(t)=\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}f(s,y(s))ds+
\int_{0}^{1}\bigg[\frac{1}{\gamma_{1}\Gamma(\alpha)\Gamma(\beta)}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}dr\nonumber\\&&-
\frac{(1-s)^{\alpha-1}}{\gamma_{1}\Gamma(\alpha)}+
\bigg(\frac{2\gamma_{2}}{\gamma_{1}\alpha\Gamma(\alpha)}-\frac{2t}
{\alpha\Gamma(\alpha)}
\bigg)(1-s)^{\alpha}\bigg]f(s,y(s))ds,\hspace{0.2cm}t\in[0,1].
\end{eqnarray}
When we put $\begin{displaystyle}\sup_{t\in[0,1]}\end{displaystyle}|f(t,0)|=M$ we show that $FB_{\rho}\subset B_{\rho},$ where
$B_{\rho}=\{y\in C([0,1],\R):\hspace{0.1cm}\|y\|\leq \rho$\} and $\rho\geq\frac{MA}{1-LA}.$\\
\\
Let $y\in B_{\rho},$ $t\in [0,1],$ this leads to write:
\begin{eqnarray}
&&\|(Fy)(t)\|\nonumber\\
&\leq&\begin{displaystyle}\sup_{t\in[0,1]}\end{displaystyle}\bigg\{
\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}f(s,y(s))ds+\frac{1}{|\gamma_{1}|\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}
\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}f(s,y(s))drds\nonumber\\
&+&\frac{1}{|\gamma_{1}|\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}f(s,y(s)) ds+\frac{2|\gamma_{2}|}{|\gamma_{1}|\alpha\Gamma(\alpha)}
\int_{0}^{1}(1-s)^{\alpha}f(s,y(s)) ds \nonumber\\
&+&\frac{2t}{\alpha\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha}f(s,y(s))ds\bigg\}\nonumber
\end{eqnarray}
\begin{eqnarray}
&\leq&\begin{displaystyle}\sup_{t\in[0,1]}\end{displaystyle}\bigg\{
\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}\bigg(|f(s,y(s))-f(s,0)|+|f(s,0)|\bigg)ds\nonumber\\
&+&\frac{1}{|\gamma_{1}|\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}
\bigg(|f(s,y(s))-f(s,0)|+|f(s,0)|\bigg)drds\nonumber
\end{eqnarray}
\begin{eqnarray}
\label{equ18}
&+&\frac{1}{|\gamma_{1}|\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}\bigg(|f(s,y(s))-f(s,0)|+|f(s,0)|\bigg)ds\nonumber\\
&+&\frac{2|\gamma_{2}|}{|\gamma_{1}|\alpha\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha}
\bigg(|f(s,y(s))-f(s,0)|+|f(s,0)|\bigg)ds\nonumber\\
&+&\frac{2}{\alpha\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha}\bigg(|f(s,y(s))-f(s,0)|+|f(s,0)|\bigg)ds\bigg\}.
\end{eqnarray}
To put $u=\frac{r-s}{1-r},$ i.e, $1-r=(1-u)(1-s),$ $dr=(1-s)du$ we obtain:
\begin{eqnarray}
\label{equ19}
\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}drds&=&\int_{0}^{1}(1-s)^{\alpha+\beta-1}ds \int_{0}^{1}(1-u)^{\beta-1}u^{\alpha-1}du\nonumber\\
&=&\frac{\mathfrak{B(\beta,\alpha)}}{\alpha+\beta}.
\end{eqnarray}
By substitution in (\ref{equ18}), and after simplification, we get
\begin{eqnarray}
\label{equ20}
\|(Fy)(t)\|&\leq&(L\rho+M)\bigg\{\frac{1}{\Gamma(\alpha+1)}+\frac{\mathfrak{B(\beta,\alpha)}}
{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+
\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}\nonumber\\
&&+\frac{2}{\Gamma(\alpha+2)}\bigg\}\nonumber\\
\\
&\leq&(L\rho+M)A\leq\rho,
\end{eqnarray}
which means that $FB_{\rho}\subset B_{\rho}.$\\
\\
Now, suppose that $x,y\in C([0,1],\R)$ and $t\in[0,1],$ then we have:
\begin{eqnarray}
\|(Fx)-(Fy)\|
&\leq&\begin{displaystyle}\sup_{t\in[0,1]}\end{displaystyle}\bigg\{
\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}|f(s,x(s))-f(s,y(s))|ds\nonumber\\
&+&\frac{1}{|\gamma_{1}|\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}
|f(s,x(s))-f(s,y(s))|drds\nonumber\\
&+&\frac{1}{|\gamma_{1}|\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}|f(s,x(s))-f(s,y(s))|ds\nonumber\\
&+&\frac{2|\gamma_{2}|}{|\gamma_{1}|\alpha\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha}
|f(s,x(s))-f(s,y(s))|ds\nonumber\\
&+&\frac{2}{\alpha\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha}|f(s,x(s))-f(s,y(s))|ds\bigg\}\nonumber
\end{eqnarray}
\begin{eqnarray}
&\leq&L\|x-y\|\bigg\{\frac{1}{\Gamma(\alpha+1)}+\frac{\mathfrak{B(\beta,\alpha)}}
{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+
\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}\nonumber\\
&&+\frac{2}{\Gamma(\alpha+2)}\bigg\}\nonumber\\
&=&LA\|x-y\|.\nonumber
\end{eqnarray}
Since we assumed $0<LA<1,$ then $F$ is a contraction. Using the principle of Banach fixed point, we deduce that the problem (\ref{equ5})-(\ref{equ7}) has a unique solution.
{\flushright$\blacksquare$\\}




%\underline{\textbf{Example:}}\\


\begin{example}\label{Example}
Consider the following boundary value problem:
\begin{eqnarray}
\label{equ21}
^CD^{\frac{3}{2}}y(t)=\frac{1}{t^{2}+4}\times\frac{|x|}{1+|x|}+t\cos^{2}t,\hspace{0.2cm} t\in [0,1]
\end{eqnarray}
\begin{eqnarray}
\label{equ22}
y(0)=\int_{0}^{1}y(s)ds
\end{eqnarray}
\begin{eqnarray}
\label{equ23}
y(1)=\frac{1}{\Gamma(\frac{1}{2})}\int_{0}^{1}(1-s)^{-\frac{1}{2}}y(s)ds.
\end{eqnarray}
In this example, $\alpha=\frac{3}{2},$ $\beta=\frac{1}{2}$ et $f(t,x)=\frac{1}{t^{2}+4}\times\frac{|x|}{1+|x|}+t\cos^{2}t.$\\
We have:
\begin{eqnarray}
|f(t,x)-f(t,y)|&=&\frac{1}{t^{2}+4}\times\frac{||x|-|y||}{(1+|x|)(1+|y|)}\nonumber\\
&\leq&\frac{1}{4}|x-y|,\nonumber
\end{eqnarray}
then, $L=\frac{1}{4}.$\\
\\
 By a simple calculation, we find: $LA\simeq 0.3715...<1,$ and by theorem \ref{Th1} we deduce that the problem (\ref{equ21})-(\ref{equ23}) has a unique solution.\\
\end{example}
\medskip


\begin{theorem}\label{Th3}{\textbf{( Arzel\`{a}-Ascoli's theorem)}}\cite{b05}
Let $A\subset C(K,\R^{n}),$  $(K=[a,b]\subset\R).$\\ $A$ is relatively compact (i.e, $\bar{A}$ is compact) if and only if:\\
1.\hspace{0.2cm} $A$ is uniformly bounded.\\
2.\hspace{0.2cm} $A$ is equicontinuous.\\
$\bullet$ Recall that a function $f$ is uniformly bounded in $A$ if there exists a constant $M>0$ such that:
$$
\|f\|=\sup_{x\in K}|f(x)|\leq M,\hspace{0.3cm}\forall \hspace{0.05cm} f\in A
$$
\end{theorem}
\begin{theorem}\label{Th4}{\textbf{(Fixed point theorem of Krasnoselskii )}}\cite{b15}
Let $F$ a non-empty set, closed and convex in a Banach space $X.$ $T_{1}$ and $T_{2}$
are two applications of $F$ in $X$ such that:\\
1.\hspace{0.2cm}$T_{1}(x)+T_{2}(y)\in F,$\hspace{0.2cm}$\forall x,y\in F,$\\
2.\hspace{0.2cm}$T_{1}$ is a contraction,\\
3.\hspace{0.2cm} $T_{2}$ is compact and continuous.\\
Then, $T_{1}+T_{2}$ has a fixed point in $F,$ i.e, there exits $x\in F$ such that $T_{1}(x)+T_{2}(x)=x.$
\end{theorem}
The second existence result is obtained by using the fixed point theorem of Krasnoselskii.
\begin{theorem}\label{Th5}
Let $f:[0,1]\times\R\rightarrow\R$ be a continuous fonctions which satisfies the conditions $(H_{1})$ and
$(H_{2}):|f(t,x)|\leq \mu(t)$,\hspace{0.2cm} $\forall (t,x)\in[0,1]$ et $\mu\in C([0,1],\R^{+}).$
Suppose that
\begin{eqnarray}
\label{equ24}
L\bigg\{\frac{\mathfrak{B(\beta,\alpha)}}
{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+
\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}+\frac{2}{\Gamma(\alpha+2)}\bigg\}<1.
\end{eqnarray}
Then, the value problem (\ref{equ5})-(\ref{equ7}) has a unique solution.
\end{theorem}
%\textbf{Proof:}\\
\subsubsection*{ \bf Proof of Theorem~\ref{Th5}}
Let $\begin{displaystyle}\sup_{t\in[0,1]}|\mu(t)|=\|\mu(t)\|.\end{displaystyle}$\\ We set $\rho^{\ast}\geq\|\mu\|\bigg\{\frac{1}{\Gamma(\alpha+1)}+\frac{\mathfrak{B(\beta,\alpha)}}
{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+
\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}+\frac{2}{\Gamma(\alpha+2)}\bigg\}$ and we consider the set
$B_{\rho^{\ast}}=\{y\in C([0,1],\R):\hspace{0.1cm}\|y\|\leq \rho^{\ast}\}.$
We define two operators  $P$ and $Q$ on $B_{\rho^{\ast}}$ by:
\begin{eqnarray}
(Py)(t)&=&\frac{1}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}f(s,y(s))ds, \hspace{0.2cm} t\in[0,1]\nonumber
\end{eqnarray}
\begin{eqnarray}
(Qy)(t)&=&\frac{1}{\gamma_{1}\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}
\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}f(s,y(s))drds\nonumber\\
&+&\frac{1}{\gamma_{1}\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1}f(s,y(s)) ds
+\frac{2\gamma_{2}}{\gamma_{1}\Gamma(\alpha+1)}
\int_{0}^{1}(1-s)^{\alpha}f(s,y(s)) ds \nonumber\\
 &+&\frac{2t}{\Gamma(\alpha+1)}\int_{0}^{1}(1-s)^{\alpha}f(s,y(s))ds, \hspace{0.2cm} t\in[0,1]\nonumber
\end{eqnarray}
Let $x,y\in B_{\rho^{\ast}},$ we write:
\begin{eqnarray}
\|Px+Qy\|&\leq& \frac{\|\mu\|}{\Gamma(\alpha)}\int_{0}^{t}(t-s)^{\alpha-1}ds+\frac{\|\mu\|}
{|\gamma_{1}|\Gamma(\alpha)\Gamma(\beta)}\int_{0}^{1}\int_{s}^{1}(1-r)^{\beta-1}(r-s)^{\alpha-1}drds\nonumber\\
&+&\frac{\|\mu\|}{|\gamma_{1}|\Gamma(\alpha)}\int_{0}^{1}(1-s)^{\alpha-1} ds+\frac{2|\gamma_{2}|\|\mu\|}
{|\gamma_{1}|\Gamma(\alpha+1)}\int_{0}^{1}(1-s)^{\alpha} ds \nonumber\\
&+&\frac{2\|\mu\|}{\Gamma(\alpha+1)}\int_{0}^{1}(1-s)^{\alpha}ds\nonumber\\
&\leq&\|\mu\|\bigg\{\frac{1}{\Gamma(\alpha+1)}+\frac{\mathfrak{B(\beta,\alpha)}}
{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+
\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}+\frac{2}{\Gamma(\alpha+2)}\bigg\}\nonumber\\
&\leq&\rho^{\ast}.\nonumber
\end{eqnarray}
Then, $Px+Qy\in B_{\rho^{\ast}},$ we have:
$$
\|Q(x)-Qy\|\leq L\|x-y\|\bigg\{\frac{\mathfrak{B(\beta,\alpha)}}
{|\gamma_{1}|(\alpha+\beta)\Gamma(\alpha)\Gamma(\beta)}+\frac{1}{|\gamma_{1}|\Gamma(\alpha+1)}+
\frac{2|\gamma_{2}|}{|\gamma_{1}|\Gamma(\alpha+2)}+\frac{2}{\Gamma(\alpha+2)}\bigg\}
$$
By exploiting (\ref{equ24}), we deduce that $Q$ is a contraction.\\
\\
According to the definition of the operator $P,$ we deduce that the continuity of $f$ implies that of $P.$ In addition, we have:
\begin{eqnarray}
\label{equ25}
\|Px\|&\leq&\|\mu\|\int_{0}^{t}\frac{(t-s)^{\alpha-1}}{\Gamma(\alpha)}ds\nonumber\\
&\leq&\frac{\|\mu\|}{\Gamma(\alpha+1)},
\end{eqnarray}
which implies that $P$ is uniformly bounded.\\
\\
Now, we show that  $P$ is compact. We have:
\begin{eqnarray}
(Py)(t_{1})-(Py)(t_{2})&=&\frac{1}{\Gamma(\alpha)}\int_{0}^{t_{1}}(t_{1}-s)^{\alpha-1}f(s,y(s))ds-
\frac{1}{\Gamma(\alpha)}\int_{0}^{t_{2}}(t_{2}-s)^{\alpha-1}f(s,y(s))ds\nonumber\\
&=&\frac{1}{\Gamma(\alpha)}\bigg(\int_{0}^{t_{1}}(t_{1}-s)^{\alpha-1}f(s,y(s))ds-
\int_{0}^{t_{1}}(t_{2}-s)^{\alpha-1}f(s,y(s))ds\nonumber\\
&-&\int_{t_{1}}^{t_{2}}(t_{2}-s)^{\alpha-1}f(s,y(s))ds\bigg)\nonumber
\end{eqnarray}
Taking into account the condition $(H_{1}),$
we set $f^{\ast}=\begin{displaystyle}\sup_{(t,x)\in[0,1]\times B_{\rho^{\ast}}}|f(t,x)|\end{displaystyle}.$ \\Then we can write:

\begin{eqnarray}
|(Py)(t_{1})-(Py)(t_{2})|&=&\frac{1}{\Gamma(\alpha)}\bigg|\int_{0}^{t_{1}}\bigg[(t_{2}-s)^{\alpha-1}-(t_{1}-s)^{\alpha-1}\bigg]
f(s,y(s))ds\nonumber\\
&&+\int_{t_{1}}^{t_{2}}(t_{2}-s)^{\alpha-1}f(s,y(s))ds \bigg|\nonumber\\
&\leq&\frac{f^{\ast}}{\Gamma(\alpha)}\bigg|\int_{0}^{t_{1}}\bigg[(t_{2}-s)^{\alpha-1}-(t_{1}-s)^{\alpha-1}\bigg]ds+
\int_{t_{1}}^{t_{2}}(t_{2}-s)^{\alpha-1}ds \bigg|,\nonumber
\end{eqnarray}
a simple calculation leads us to:
\begin{eqnarray}
\label{equ26}
|(Py)(t_{1})-(Py)(t_{2})|&\leq&\frac{f^{\ast}}{\Gamma(\alpha+1)}|t_{2}^{\alpha}-t_{1}^{\alpha}|.
\end{eqnarray}
The second member of (\ref{equ26}) is independent of $y$ and tends to zero when $t_{2}-t_{1}\rightarrow 0,$ so $P$ is equicontinuous. Using the Arzel\`{a}-Ascoli theorem, we deduce that  $P$ is compact in $ B_{\rho^{\ast}}.$ Thus all the assumptions of the fixed point theorem of Krasnoselskii are satisfied. Which implies that the boundary value problem (\ref{equ5})-(\ref{equ7}) has a unique solution on  $[0,1].$
\newpage

%\noindent\textsc{B.Tellab } \\
%Department of Mathematics and Informatics, Ghardaia University \\
 %47000 Ghardaia, Algeria  \\
%e-mail:   brahimtel@yahoo.fr \smallskip

%\noindent\textsc{K.Houam }\\
%Department of Mathematics and Informatics, University of Tebessa \\
% 12000 Tebessa, Algeria\\
%e-mail: haouam@yahoo.fr\\
\end{document} 